Author's response to reviews

Title: Mapping Turnaround Times (TAT) to a Generic Timeline: A Systematic Review of TAT Definitions in Clinical Domains

Authors:

Bernhard Breil (bernhard.breil@ukmuenster.de)
Fleur Fritz (fleur.fritz@ukmuenster.de)
Volker Thiemann (volker.thiemann@ukmuenster.de)
Martin Dugas (martin.dugas@ukmuenster.de)

Version: 4 Date: 12 May 2011

Author's response to reviews: see over
Dear Sir or Madam,

Hereby I submit the revised manuscript "Mapping Turnaround Times (TAT) to a Generic Timeline: A Systematic Review of TAT Definitions in Clinical Domains" for publication in BMC Medical Informatics and Decision Making.

We updated the remaining 4 TAT abbreviations as suggested by referee 2 and reformatted the paper as specified in the BMC authors checklist.

I am looking forward to hearing about the acceptance of the revised manuscript.

Best regards,

Bernhard Breil